Author's response to reviews

Title: Effect of the rs2259816 polymorphism in the HNF1A gene on circulating levels of C-reactive protein and coronary artery disease (The Ludwigshafen Risk and Cardiovascular Health Study)

Authors:

   Marcus E Kleber (marcus.kleber@synlab.de)
   Tanja B Grammer (tania.grammer@synlab.com)
   Wilfried Renner (rennerw@klinikum-graz.at)
   Winfried März (maerz@synlab.com)

Version: 4 Date: 5 November 2010

Author's response to reviews:

Dear editors,

I included the tables in the main manuscript and removed the additional WORD document.
I updated the abstract in the submission system.
I changed the red color of the updated text passages to black.
I removed the additional cover letter.
I changed the style of the reference list to include all authors up to 30. I also removed the duplicated reference list.
Finally, I checked the typography.

Kind regards

Marcus Kleber